MOSS Meeting Minutes

October 28, 2003
ELEMENT ATTENDANCE:

ESDIS SSDO and System Manager, MRTT, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar

NEXT TELECON: Debrief – November 4, 2003 @ 2pm EST – same dial in information

USA Toll Free Number: 888-395-8106 

PASSCODE:  92854

International dial-in 

USA Toll Number: +1-484-630-6975
MEETING HIGHLIGHTS:

Agenda

1. MOSS 3 Elements Status

MLS

Audrey mentioned that they have a couple of things outstanding: QA and DAP; they planned to try to complete today. Their efforts have been concentrating on L3s and PH. In summary: received all L0 data; processed and delivered L1B for DOY 108; had known problem with DOY 107; produced and delivered all L2s with PH except ML2T (metadata problem), ML2RHI (not created - configuration problem), and ML2DGG and ML2DGM (not MC - turned off for performance reasons); output delivered for L3s daily and monthly but did not ingest into the DAAC and is being investigated; and completed integrated search/order for MTMGW. 

TES

Doug reported that a disk went down during ftp transfer of some L1Bs to ASDC DAAC. They halted processing of day 2 global survey data; transferred all L2s to ASDC and partially the L1Bs. Once the disk is back up, ASDC will like for TES to try to send the remaining L1B products. Andy mentioned that they have received all of their L0 data and at least one granule for each ESDT has been delivered to ASDC. They had experience both hardware and software problems during the test.

HIRDLS

Vince mentioned that they still have BROWSE to do. They produced and delivered L1 products for all 3 days and processed up to L2 with PH and QA for one day. They have received all of their L0 data. He confirmed that their PDRs were automatically generated by the system.

OMI

OMI reported that they have produced and delivered all of their L1B products to the DAAC; some L2s did not go into the DAAC and will re-run on Thursday (10/30/03) during the Engineering Test with GDAAC and ODPS; and have performed a search with MTMGW and are going to do an order. Elaine reported that they received all of their L0 data that were expected; three L1Bs failed ingest into the DAAC due to no NOSE parameters generated (1 IRR and 2 zoomed products); two L2s (OMDOAO3 (not MC) and OMCLDO2) were not successfully ingested by the DAAC and two L2s (OMTO3 and OMNO2) were ingested okay.

EDOS

Ron reported that for an EDOS perspective the test went well. They wrote 4 DRs. The Oracle problems observed during last MOSS did not occur and there were not problems with EDOS supporting the 4 missions concurrently.

MRTT

Susan reported that the ground stations did okay for this test. They do anticipate during more ground testing with them outside of MOSS 4. They are working on completing verification of EMOS archival/retrieval of history data objectives/requirements.

Averstar

Dot reported that according to Scott, he has ordered everything in the archive except MLS L3s and OMTO3. Karen mentioned that OMTO3 should be able to be ordered now. MLS L3s did not get ingested into the DAAC. Dot was going to check if he were able to order QA from HIRDLS and PH from MLS.

2. ASDC/TES SIPS Network

Karen mentioned that the only change that she was aware of that corrected the slow transfer problem during the MOSS test was that the MISR ftp window size was adjusted. Steve mentioned that it was changed from 240 to 256. Doug is interested in more testing of the network to be performed after MOSS 3 with TES and MISR participating. Doug asked who would lead up that effort. Karen mentioned that she would try to coordinate something for next week if all is interested. Steve added that he would prefer to monitor MISR and the network for the remainder of the week to see if the current configuration is the correct one for MISR and then add components to the testing one by one. Andy mentioned that when the threads were set to 5 for MISR and 5 for TES, they got all of their data and experienced some timeouts earlier in test but on Thursday and Friday of last week everything was good. Andy recommended, to be fair, the threads should be 3 for TES and 3 for MISR. Glenn Iona suggested that he and Karen put together an outline detailing what type of testing to be done and circulate it to the appropriate persons.

3. MOSS 4 Test Data

All new test data for MOSS 4 are due to ESDIS no later than December 12th.  Please set any quick look flags prior to delivery. TES and OMI have indicated that they will have new test data for MOSS 4. TES expects to deliver ahead of schedule. OMI would like to use Mission Test #2 (MT #2) data. Karen asked that they come up with a fall back plan in case the MT#2 data is not available in time for MOSS 4. HIRDLS and MLS have to relay their plans to Karen. MLS indicated that they would like to get all of DOY 106 data if possible. ESDIS will check to see if possible. Karen mentioned that GBAD would be re-run for MOSS 4, such that the data would be continuous. This would enable processing of DOY 107 data as well.
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